Author’s response to reviews

Title: Erythropoietin, uncertainty principle and cancer related anaemia

Authors:

Otavio A C Clark (clark@evidencias.com)
Jared R Adams (j-adams4@nwu.edu)
Charles L Bennet (cbenne@northwestern.edu)
Benjamin B Djulbegovic (djulbebm@moffitt.usf.edu)

Version: 3 Date: 17 Sep 2002

We addressed all the requests from the referee:

1) We run TWO meta-regressions and decided to cite both - one with no significant result and another wit(see the text)

2) The values for RR - all were reviewed and one corrected that was mistyped in the text

3) The changes in the text suggested

4) An explanation about heterogeneity when excluding the Porter study

We also formatted the text as requested by the editorial committee: